Author's response to reviews

Title: Surveillance study of vector species on board passenger ships. Risk factors related to infestations

Authors:

Varvara A Mouchtouri (mouchtourib@med.uth.gr)
Rimma Anagnostopoulou (ranagnost@med.uth.gr)
Anna Samanidou-Voyadjoglou (asaman@otenet.gr)
Kalliopi Theodoridou (ktheod@med.uth.gr)
Chrissi Hatzoglou (chatz@med.uth.gr)
Jenny Kremastinou (jkrem@nsph.gr)
Christos Hadjichristodoulou (xhatzi@med.uth.gr)

Version: 3 Date: 13 March 2008

Author's response to reviews:

Larissa March 13, 2008

To: The Editor
BMC Public Health

Dear Editor,

We would like to thank you very much for accepting the article and to submit the final version of the research article entitled "Surveillance study of vector species on board passenger ships. Risk factors related to infestations."

Yours faithfully,

Christos Hadjichristodoulou, MD, PhD
Assistant Professor
Medical Faculty
Department of Hygiene and Epidemiology
22 Papakyriazi Street
41222 Larissa
Greece
Tel: +30 2410 565259
Fax: +30 2410 565259
E-mail: xhatzi@med.uth.gr